Author's response to reviews

Title: Developing a facilitation model to promote organisational development in primary care practices

Authors:

Melody Rhydderch (RhydderchM@Cardiff.ac.uk)
Adrian Edwards (edwardsag@Cardiff.ac.uk)
Martin Marshall (martin.marshall@manchester.ac.uk)
Glyn Elwyn (elwyn@cardiff.ac.uk)
Grol Richard (r.grol@kwazo.umcn.nl)

Version: 4 Date: 5 April 2006

Author's response to reviews:

Dear Alex

Thank-you for accepted my amended submission. Please find attached a resubmitted version of the manuscript that takes account of your request to format the article in the BMC style. I am writing to confirm that the details of ethical approval are that it was obtained on 16th May 2003. The application was titled 'European Practice Assessment: improving management in primary care'.

I also corrected my references with endnote turned off but when I submitted I can see that ref 12 and 24 did not correct themselves. Just to confirm they should read as follows:


The number is MREC 03/9/032.

Please let me know if you need anymore information.

Kind regards

Melody Rhydderch